Author's response to reviews

Title: Assessment of phytochemicals, antioxidant, anti-lipid peroxidation and anti-hemolytic activity of extract and various fractions of Maytenus royleanus leaves

Authors:

Maria Shabbir (comrademaria@gmail.com)
Muhammad Rashid Khan (mrkhanqau@yahoo.com)
Naima Saeed (naima_qau@hotmail.com)

Version: 2 Date: 22 January 2013

Author's response to reviews:

Dear Editor

We have made/corrected the following changes in the manuscript.

Your cover letter would seem to imply that this manuscript represents a resubmission of a manuscript that has been previously submitted to the journal. However, we are unable to find any record of a manuscript from yourselves dealing with studies on this particular plant. If this does represent a resubmission, could you please provide us with the manuscript ID number of the original submission?

This manuscript has not been submitted previously to this journal. The cover letter could be a copy of some previous one.

We note that there is a degree of overlap between the text of your current submission, and that of a number of previously published papers from yourselves. This overlap is mostly located in the methods section. Could we please ask you to revise your manuscript to try and limit this overlap as much as possible.

We have revised Materials and Methods section

With best wishes

Dr Muhammad Rashid Khan

Associate Professor